

Author's response to reviews

Title: Younger Age of Escalation of Cardiovascular Risk Factors in Asian Indian Subjects

Authors:

Rajeev Gupta (rajeevg@satyam.net.in)
Anoop Misra (anoopmisra@gmail.com)
Naval K Vikram (navalvikram@gmail.com)
Dimple Kondal (navalvikram@gmail.com)
Shaon Sen Gupta (navalvikram@gmail.com)
Aachu Agrawal (jaipur_agrawal@yahoo.com)
Ravindra M Pandey (rmpnadey@yahoo.com)

Version: 3 **Date:** 2 June 2009

Author's response to reviews: see over

To
The Editor
BMC Cardiovascular Disorders.
www.biomedcentral.com

Ref:MS: 1897871095245849. Younger age of escalation of cardiovascular risk factors in Asian Indian subjects.

Sir:

The manuscript has been revised in view of the comments of the editors and referees. The following changes have been made:

1. Statement conforming informed consent has been added to the methods section.

Referee 1:

2. Legends of tables and figures have been changed.
3. In figures, clear contrasting colours and lines have been added.

Referee 2:

4. The introduction and discussion has been edited.

Thanks.

Yours faithfully,

Dr Rajeev Gupta.